CS148 Building Intelligent Robots

Standard Track Enrollment Form 07 Sep 2004

Name: Grad Year: Email: Concentration:

Enrollment in the Standard Track of CS148 is limited to 26 students due to the limited number of LEGO Mindstorm robot kits. It is therefore important that each student be dedicated to the course and put forth significant effort throughout the semester. Completing this form will enter you into a lottery for a spot in the course. Completed forms must be submitted to Chad Jenkins (CIT 327) by 11:59pm on Thursday, September 9th. Please only enter the lottery if you can make a sincere commitment. We will tell you via email whether or not you have been selected. If you do not hear from us by noon on Sunday, email cs148tas@cs.brown.edu.

Why do you want to take this course? What are your expectations?

Please describe any experience you have had with robots or with the LEGO Mindstorms kits. Also list any other relevant courses you have taken at Brown, and describe why that background will be helpful here. If you have not taken CS4, CS15/16, or CS17/18, please describe any relevant programming experience.